Author’s response to reviews

Title: Understanding Resilience of Female Adolescents towards Teenage Pregnancy: A Cross-sectional Survey in Dar es Salaam, Tanzania

Authors:

Constanze Pfeiffer (constanze.pfeiffer@unibas.ch)

Collins Ahorlu (collins.ahorlu@gmail.com)

Sandra Alba (S.Alba@kit.nl)

Brigit Obrist (brigit.obrist@unibas.ch)

Version: 2 Date: 08 Dec 2016

Author’s response to reviews:

REPH-D-16-00080

Understanding Resilience of Female Adolescents towards Teenage Pregnancy: A Cross-sectional Survey in Dar es Salaam, Tanzania

Constanze Pfeiffer, Ph.D.; Collins Ahorlu, PhD; Sandra Alba, PhD; Brigit Obrist, Professor
Reproductive Health

Dear Venkatraman Chandra-Mouli

We thank you for your interest in our manuscript, and for providing such helpful comments and suggestions. An English native speaker has now proof-edited the manuscript. Thus we believe that the quality of the manuscript could further be improved.
We hope that you are willing to consider our manuscript for publication in Reproductive Health.

Sincerely,

Constanze Pfeiffer